Author's response to reviews

Title: A comparative examination of tuberculosis immigration medical screening programs from selected countries with high immigration and low tuberculosis incidence rates

Authors:

Gonzalo G Alvarez (galvarez@ohri.ca)
Brian Gushulak (brian.gushulak@gmail.com)
Khaled A Rumman (tbjor@go.com.jo)
Ekkehardt Altpeter (ekkehardt.altpeter@bag.admin.ch)
Daniel Chemtob (daniel.chemtob@moh.health.gov.il)
Paul Douglas (Paul.Douglas@immi.gov.au)
Connie Erkens (ErkensC@kncvtbc.nl)
Peter Helbling (peter.helbling@bag.admin.ch)
Ingrid Hamilton (Ingrid_Hamilton@moh.govt.nz)
Jane Jones (Jane.Jones@HPA.org.uk)
Alberto Matteelli (amatteelli@bsnet.it)
Marie-Claire Paty (marie-claire.paty@sante.gouv.fr)
Drew Posey (dposey@cdc.gov)
Daniel Sagebiel (SagebielD@rki.de)
Erika Slump (SlumpE@kncvtbc.nl)
Anders Tegnell (Anders.Tegnell@socialstyrelsen.se)
Elena Rodríguez Valín (ervalin@isciii.es)
Brita Winje (Brita.Askeland.Winje@fhi.no)
Edward Ellis (Edward_Ellis@phac-aspc.gc.ca)

Version: 7 Date: 12 November 2010

Author's response to reviews:

We made two changes:

1) We added Daniel Sagebiel from Germany as an author as we finally tracked him down at his new post in Manila Philippines and he has agreed to the contents of the manuscript and sent in his copyright form

2) We added the suggested comment regarding Dr. Miyano, "Dr. S. Miyano contributed to the acquisition of the data and as such we feel he is entitled to authorship but, despite our extensive efforts, we were unable to contact him to approve the final content of this manuscript and his authorship. Therefore, we acknowledge his contribution here."